Author’s response to reviews

Title: Developing Outcome, Process and Balancing Measures for an Emergency Department Longitudinal Patient Monitoring System using a Modified Delphi

Authors:
Marie Elizabeth Ward (m.ward@ucd.ie)
Abel Wakai (AWakai@rcsi.ie)
Ronald McDowell (ronaldmcdowell@rcsi.ie)
Fiona Boland (fionaboland@rcsi.ie)
Eoin Coughlan (eoincoughlan@yahoo.com)
Moayed Hamza (Moayed.hamza@ucdconnect.ie)
John Browne (j.browne@ucc.ie)
Ronan O'Sullivan (rgosullivan@bonsecours.ie)
Una Geary (ugeary@stjames.ie)
Fiona McDaid (fiona.mcdaid@hse.ie)
Éidin Ni Shé (eidin.nishe@ucd.ie)
Frances Drummond (francesjdrummond22@gmail.com)
Conor Deasy (Conor.Deasy@hse.ie)
Eilish McAuliffe (Eilish.mcauliffe@ucd.ie)

Version: 4 Date: 21 Dec 2018

Author’s response to reviews:

Dear Editor,

Appendices 1 and 2 have now been uploaded and labelled as requested and an additional section has been added in the main manuscript listing these files.
Many thanks,

Eilish McAuliffe